Author's response to reviews

Title: Ontario primary care reform and quality improvement activities: an environmental scan

Authors:

Charmaine McPherson (cmcpherso@stfx.ca)
Shannon Sibbald (ssibbald@uwo.ca)
Anita Kothari (akothari@uwo.ca)

Version: 2 Date: 7 August 2012

Author's response to reviews:

The methods section has been revised to include a more specific statement about ethics review on page 7, stating:

The St. Francis Xavier University Research Ethics Board (REB) deemed the project exempt from ethics review given that participants were describing characteristics of programs.